Author's response to reviews

Title: Social factors associated with centenarian rate in 32 OECD countries

Authors:

Jong In kim (kji122@wku.ac.kr)

Version: 2 Date: 21 September 2012

Author's response to reviews: see over
Social factors associated with centenarian rate in 32 OECD countries

Jong In, Kim
Division of Health and Welfare, Wonkwang University, Republic of Korea

Corresponding Author: Jong In, Kim
Professor and Head, Division of Health and Welfare, Wonkwang University, 570-749, Republic of Korea Email: kji122@wku.ac.kr